Author's response to reviews

Title: Subanesthetic ketamine for the treatment of children and adolescents with chronic pain: a longitudinal study

Authors:

Kathy A. Sheehy (KSheehy@childrensnational.org)
Elena Muller (emuller1@tulane.edu)
Caroline Lippold (CLippold@childrensnational.org)
Mehdi Nouraie (snouraie@howard.edu)
Julia C Finkel (JFINKEL@childrensnational.org)
Zenaide M Quezado (zquezado@cnmc.org)

Version: 4 Date: 11 November 2015

Author's response to reviews: see over
Response to Reviewers and to the editor

Reviewer 2

1. I would just suggest to add in the section ‘methods’ information regarding the study design (retrospective, observational study)

Response: We have added the study design to the methods (Page 6, lines 6-7)

Additional formatting request:

1. Please clarify in the Authors Contributions that all authors read and approved the final version of the manuscript.

Response: We have revised the text as suggested. Page 17 line 20

2. List of abbreviations:
If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations can be provided, which should precede the competing interests and authors’ contributions.

Response: A list of abbreviations was added to page 17 lines 1-6.